\magnification=\magstep1
\nopagenumbers
\parindent=0pt
{\bf Math 226: Calculus IV \hfill Spring 1992}


\centerline{\bf Course Information}


\bigskip
TEXTS:
Shaum's Outline Series: Matrix Operations, Chapters 1, 2, 4, 5, 6, 11.

Boyce \& DiPrima, Elementary Differential Equations and
Boundary Value Problems, {\sl Fifth Edition}, Chapters 1--3, 5 with some sections omitted.


\bigskip
EXAM SCHEDULE:  There will be three one-hour exams and a
two hour final on the following dates:
$$\vbox{\halign{#\
\hfil&#\hfil\cr Exam I &Thursday, February 13, 8:00 A.M.\cr
Exam II &Thursday, March 19, 8:00 A.M.\cr
Exam III &Tuesday, April 21, 8:00 A.M.\cr
Final &Tuesday, May 5, 7:30 P.M.\cr
}}$$
The hour exams will be held in Stepan Center. Exams may be made up
only with an  excused absence from the Associate Vice
President for Residence Life. Conflicts with exams in
other courses must be resolved during the first week of
classes. Quizzes will be given periodically in class.
 
\bigskip
TUTORIALS:  There will be three tutorials per week:
$$\vbox{\halign{#\ \hfil&#\hfil\cr
Tuesday 3:00--5:30 P.M. &CCMB 226\cr
Wednesday 7:00--9:00 P.M. &CCMB 226\cr
Thursday 3:00--5:30 P.M. &CCMB 226\cr
}}$$
The tutorials will be staffed by Teaching Assistants who
will answer students' questions either individually or
in small groups. You may attend any tutorial you wish. 
Graded exams and quizzes may be picked up
in any of the tutorials but {\sl only during the week
following the exam or quiz}.

\bigskip
OFFICE HOURS: The offices are in the Mathematics Building:
$$\vbox{\halign{#\ \hfil&#\hfil\ &#\hfil\cr
Dennis Snow &203&MW 3:30--4:30 P.M.\cr
\v Zarko Bi\v zaca &218& MW 2:30--3:30 P.M.\cr
\v Zeljko Sokolovic &304& F 2--4 P.M.\cr
Kirk Weller &316& Th 9--11 A.M.\cr
}}$$


\bigskip
GRADES:  Grades will be based on a total of 500 points broken down as follows:
$$\vbox{\halign{#\ \hfil&#\hfil\cr
Exam I--III &100 each, 300 total\cr
Final &150\cr
Extra &50\cr
}}$$
The extra 50 points will be based on quizzes and optional
assignments. Current grades will be mailed regularly to
each student's computer account.


\bye
